Author’s response to reviews

Title: Behavioural Sequelae Following Acute Kawasaki Disease

Authors:

Daniel Carlton-Conway (Robert.Tulloh@gstt.nhs.uk)
Raju Ahluwalia (Robert.Tulloh@gstt.nhs.uk)
Lucy Henry (Lucy.Henry@kcl.ac.uk)
Colin Michie (colinm@easynet.co.uk)
Robert Tulloh (Robert.Tulloh@gstt.nhs.uk)

Version: 2 Date: 9 February 2005

Author’s response to reviews:

Cover letter for submission of paper

Title: Behavioural sequelae following acute Kawasaki disease
Authors: Daniel Carlton-Conway1, Raju Ahluwalia1, Lucy Henry3, Colin Michie2, Robert Tulloh1

Dear Sir

I enclose our revised version of the above paper. We have taken note of the reviewers comments and enclose our comments individually

1. The reviewer noted that the Figure 1 was rather better labeled as a table. We have therefore called it Table 1 and altered the subsequent numbering accordingly
2. The reviewer pointed out the spelling mistake in Lymphadenopathy which we have now corrected in Table 1
3. The reviewer commented that the PET scan marker should be described and this has now been done.
4. The formatting for instructions for authors has been carefully read and a few minor changes have been made

We hope that you are satisfied with the changes and look forward to publication in your journal.

With regards

Dr Robert Tulloh